Author’s response to reviews

Title: Homozygous NOTCH3 p.R587C mutation in Chinese Patients with CADASIL: a case report

Authors:

Ruojie He (herj@mail2.sysu.edu.cn)
Huan Li (lihuan26@mail2.sysu.edu.cn)
Yiming Sun (13570397296@163.com)
Menglong Chen (cmlcalm@sina.com)
Liang Wang (lion.wong@foxmail.com)
Yuling Zhu (zhouxs1011@163.com)
Cheng Zhang (chengzhang100@foxmail.com)

Version: 2 Date: 16 Feb 2020

Author’s response to reviews:

Feb 16, 2020

Dr. Darren Byrne,
Editor-in-Chief
BMC Neurology

Dear Dr. Darren Byrne,

Thank you very much for giving us an opportunity to revise our manuscript. We appreciate the editor and reviewers very much for their positive and constructive comments and suggestions on our manuscript entitled “Homozygous NOTCH3 p.R587C mutation in Chinese Patients with CADASIL: a case report” (NURL-D-19-00776R2). Those comments, which are of great guiding significance to our study, are valuable and helpful for revising and improving the quality of our paper.

We have studied comments carefully and made correction which we hope meet with approval. The main corrections in the paper and the responds to the editor and reviewer’s comments are as following:

1. We have added the heading "Declarations" before Declarations subheadings list.

2. In the Funding section, we have stated clearly the role the funding.

3. We have uploaded Figures 1 to 3 in the submission system.

4. In the Abstract, Introduction has been renamed to Background.

5. In the Body of Manuscript, Introduction has been renamed to Background.
We have tried our best to improve the manuscript and made changes in the manuscript. We appreciate for Editor and Reviewers’ warm work earnestly, and hope that the correction will meet with approval.

Once again, thank you very much for your comments and suggestions.

Cheng Zhang, MD, PhD
Department of Neurology, First Affiliated Hospital,
Sun Yat-Sen University, No. 58 Zhongshan 2nd Road, 510080, Guangzhou, Guangdong, China
Tel.: +86 20 87755766
Fax: +86 20 87750632
E-mail: chengzhang100@foxmail.com